%
% Fill only lines marked with dots  
% 
%
%
\documentclass[10pt,a4paper,twoside]{article}
\usepackage{latexsym,amssymb}
\usepackage{amssymb,amsmath}
\usepackage{graphicx}
\usepackage{color}
\usepackage{fancyhdr}
\pagestyle{fancy}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%  PLEASE, ONLY IN ENGLISH                                         %
%  PLEASE, ONLY ONE PAGE                                           %
%  PLEASE, USE LATEX2E                                             %
%                                                                  %
%  PLEASE, DO NOT DEFINE NEW COMMANDS                              %
%  USE NEWTHEOREMS BELOW, PLEASE, IF YOU WRITE THEOREMS AND SUCH   % %                                                                  %
%  PLEASE, CONTROL THE SIZE OF YOUR ABSTRACT BEFORE YOU SEND IT    %
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
\newtheorem{thm}{Theorem}
\newtheorem{defi}[thm]{Definition}
\newtheorem{axi}[thm]{Axiom}
\newtheorem{clm}[thm]{Claim}
\newtheorem{col}[thm]{Corollary}
\newtheorem{lem}[thm]{Lemma}
\newtheorem{con}[thm]{Conjecture}
\newtheorem{hyp}[thm]{Hypothesis}
\newtheorem{rem}[thm]{Remark}
\newcommand{\proof}{\noindent{\bf Proof:}\hspace{0.2cm}}
\setcounter{thm}{0}
\newcommand{\N}{{\mathbb N}}
\newcommand{\R}{{\mathbb R}}
\newcommand{\W}{W_0^{1,p}(\Omega)}
\newcommand{\eps}{{\varepsilon}}
\newcommand{\ds}{\displaystyle}

\newcommand{\F}{\mathcal F}
\newcommand{\E}{\mathscr E}
\newcommand{\J}{\mathscr{J}}
\newcommand{\Pl}{(\mathscr{P})}
\newcommand{\CC}{\mathbb C}
\newcommand{\into}{\int\limits_{\Omega}}
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\textwidth 15.0cm
\textheight 22.0cm
\oddsidemargin 0.4cm
\evensidemargin 0.4cm
\topmargin  0.0cm

\frenchspacing

\headwidth=\textwidth 	
\fancyhead[HCO]{9th Int. Conf. on Mathematics and Informatics,
	September 7-8, 2023, T\^argu Mure\c s, Romania}
\lhead{}\rhead{}
\fancyfoot{}
%


\begin{document}
\begin{thebibliography}{9} 
% Please, use the following patterns
%
% for articles 

\bibitem {Cat07}T. C\u{a}tina\c{s}, The bivariate Shepard operator of
Bernoulli type, \textit{Calcolo} \textbf{44}, 4 (2007), 189–-202.
%------------------------------------------------
\bibitem{CatMal22}T. Cătinaș, A. Malina, The combined Shepard operator of inverse quadratic and inverse multiquadric type, \textit{Stud. Univ. Babeș-Bolyai Math.} \textbf{67} (2022), 579–-589.
%------------------------------------------------
\bibitem {FraNie80}R. Franke, G. Nielson, Smooth interpolation of large sets
of scattered data, \textit{Int. J. Numer. Meths. Engrg.}  \textbf{15} (1980), 1691–-1704. 
%------------------------------------------------
\bibitem{Pae22}P. Paewpolsong et al., Image reconstruction by shapefree radial basis function neural networks (RBFNs), \textit{Journal of Physics: Conference Series 2312.1}
(2022).
%------------------------------------------------
 \bibitem{She68}D. Shepard, A two dimensional interpolation function for irregularly spaced data, \textit{Proc. 23rd Nat. Conf. ACM} (1968), 517–-523.
 %------------------------------------------------
\bibitem{UhSka05}K. Uhlir, V. Skala, Reconstruction of damaged images using Radial Basis Functions, \textit{13th European Signal Processing Conference} (2005), 1–-4.

\end{thebibliography}


\begin{center}{\Large\bf
An application of Shepard operator in image reconstruction
}\end{center}
%
\begin{center}{\large\bf\noindent
Andra Malina
}\\[2mm]
%%
Faculty of Mathematics and Computer Science, Babeș-Bolyai University \& \\ Tiberiu Popoviciu Institute of Numerical Analysis, Romanian Academy
%%
\\[1mm]\texttt{
%%	
andra.malina@ubbcluj.ro
%%
}\end{center}
%

\vspace*{7mm}
Shepard's method, introduced in \cite{She68}, is one of the best ways to solve scattered data approximation problems, i.e., to reconstruct an unknown functions' values from some given scattered data. Multiple improvements were made during the time to obtain better results than the ones provided by the original method (e.g., \cite{Cat07}, \cite{FraNie80}).
 One example of improvement is the combination of the bivariate Shepard operator with inverse quadratic and inverse multiquadric radial basis functions which was introduced in \cite{CatMal22}.
\par Besides function approximation, another topic that uses Shepard method is the one of image reconstruction. With some ideas from \cite{Pae22} and \cite{UhSka05}, we will focus on reconstructing damaged black and white or color images, using the combined Shepard operator of inverse quadratic and inverse multiquadric type and evaluate the results by studying the approximation errors.   

%

\end{document}


